Author's response to reviews

Title: Evaluation of the utilization of the preanaesthetic clinics in a University Teaching Hospital

Authors:

Seetharaman Hariharan (uwi.hariharan@gmail.com)
Deryk Chen (uwideryk@yahoo.com)
Lorna Merritt-Charles (lmerritt@tstt.net.tt)

Version: 5 Date: 18 May 2006

Author's response to reviews:

The formatting changes required by the corepreaccept editorial team has been made in the manuscript. In the abstract, the submission system is not allowing percentage symbols, hence the symbol had to be spelt out.